Author's response to reviews

Title: School-based mental health intervention for children in war-affected Burundi: a cluster randomized trial

Authors:

Wietse A Tol (wtol@jhsph.edu)
Ivan H Komproe (ivan.komproe@hntpo.org)
Mark JD Jordans (mark.jordans@hntpo.org)
Aline Ndayisaba (ndayisaba_aline@yahoo.fr)
Prudence Ntamutumba (prunta70@gmail.com)
Heather Sipsma (heather.sipsma@yale.edu)
Eva Smallegange (esmallegange@yahoo.com)
Robert D Macy (RobertMacy@comcast.net)
Joop TVM de Jong (jtvmdejong@gmail.com)

Version: 2
Date: 31 January 2014

Author's response to reviews:

Please see the uploaded document